Author’s response to reviews

Title: Burden of varicella in Latin America and the Caribbean: findings from a systematic literature review

Authors:

Luiza Falleiros Arlant (luizahfc@terra.com.br)
Maria Pirez Garcia (mcpirez@yahoo.com)
Maria Aguero (avilaaguero@gmail.com)
Miguel Cashat (miguel_cashat@merck.com)
Cintia Parellada (cintia.parellada@merck.com)
Lara Wolfson (submissions@jssresearch.com;lara.wolfson@merck.com)

Version: 2 Date: 27 Mar 2019

Author’s response to reviews:

Natalie Pafitis, Editor
Eva Szunyogova, Editor
BMC Public Health
27 March 2019

Dear Editors

Manuscript ID: PUBH-D-18-04412R1

Manuscript title: Burden of varicella in Latin America and the Caribbean: findings from a systematic literature review.

Thank you for the additional reviewer comments on our manuscript. As requested, we are pleased to submit a clean version of the revised manuscript and a point by point response to the Assistant Editor comments.
We confirm that the map depicted in Figure 1 was drawn by the studio at PAREXEL and is not taken from another source.

Thank you very much

Yours sincerely

Lara J. Wolfson, PhD
Executive Director, Biostatistics and Research Decision Sciences (BARDS)
Merck & Co., Inc., Kenilworth, NJ 07033, USA
Tel: +1.267.305.0650
Fax: +1.862.214.3622
Email: Lara.wolfson@merck.com

LAC Varicella Manuscript – Response document: PUBH-D-18-04412R1

Assistant Editor Comments:

1. Authors Contributions

-- Please represent authors' names using their full initials, not their full name, in the Authors’ Contributions section. If there are any duplicated initials, please differentiate them to make it clear that the initials refer to separate authors.

-- Please include a statement in the Authors' contributions section to the effect that all authors have read and approved the manuscript, and ensure that this is the case.
Author reply: The changes requested have been made in the Authors’ Contributions section. We confirm that all authors have read and approved the manuscript.

2. Map Attribution

-- Please state in the cover letter whether the map depicted in figure 1 is your own or taken from another source. If taken from another source please acknowledge the source in the figure legend, and if it is under copyright also state the written permission given to use and adapt it.

If the above conditions are not met the image needs to be removed. Please note the editors may request proof of permission at any time.

Should you require an alternative source you may wish to try Wikimedia Commons:

http://commons.wikimedia.org/wiki/Main_Page

Author reply: We have added a note in the cover letter to confirm that the map depicted in Figure 1 was drawn by the studio at PAREXEL and is not taken from another source.

3. Availability of Data and Materials

Please change this to all data analysed is contained within the manuscript and its additional files.

Author reply: This change is now made.

4. Additional Files

-- Please also add a section “Additional files” where you list the following information about your supplementary material:
- File name
- Title of data
- Description of data

Author reply: These details are added on page 28.

Please ensure that all additional files are explicitly referred to in the main text.

Author reply: We confirm that all the Additional files are referred to in the main text.

5. PRISMA

Please remove this from your additional files as it is no longer required at this stage in the editorial process.

Author reply: The PRISMA document is now removed from the uploaded documents.

6. Clean Manuscript

At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.

Author reply: Thank you – the manuscript is uploaded as a single, clean version, as requested. All figures, tables, and supplementary files are cited in the manuscript text.